%Title: Five-loop vacuum polarization in pQCD: O(m_q^2\alpha_s^4 n_f^2) contribution 
%Author: P.A. Baikov, K.G. Chetyrkin and J.H. K\"uhn
%Published: <EM> Phys.Lett.B </EM> <B> 559 </B> (2003) 245-251. 
%hep-ph/0212303

\documentclass[12pt]{article}

%\setlength{\overfullrule}{5pt}

\setlength{\textwidth}{14.5cm}
\setlength{\textheight}{22.5cm}
%\setlength{\oddsidemargin}{3.cm}
\setlength{\oddsidemargin}{1.cm}
%\setlength{\headheight}{0.5cm}
\setlength{\headsep}{1.0cm}
\setlength{\topmargin}{0.0cm}
\setlength{\topskip}{0.1cm}
%\setlength{\footheight}{0.5cm}
\renewcommand{\baselinestretch}{1.1}
%\setlength{\changebarsep}{0mm}
%%%\setlength{\mathindent}{0mm}
%\setlength{\overfullrule}{5pt}
\setlength{\footskip}{1.5cm}
\frenchspacing

%\pagestyle{myheadings}
%\newcommand{\textbf}[1]{{\bf #1}}  

\newcommand{\ep}{\epsilon}
\newcommand{\la}{\lambda}
\newcommand{\La}{\Lambda}
\newcommand{\prd}{\partial}

\setlength{\textwidth}{14.5cm}
\setlength{\textheight}{22.5cm}
%\setlength{\oddsidemargin}{3.cm}
\setlength{\oddsidemargin}{1.cm}
%\setlength{\headheight}{0.5cm}
\setlength{\headsep}{1.0cm}
\setlength{\topmargin}{0.0cm}
\setlength{\topskip}{0.1cm}
%\setlength{\footheight}{0.5cm}
\renewcommand{\baselinestretch}{1.1}
%\setlength{\changebarsep}{0mm}
%%%\setlength{\mathindent}{0mm}
%\setlength{\overfullrule}{5pt}
\setlength{\footskip}{1.5cm}
\frenchspacing

%\pagestyle{myheadings}
%\newcommand{\textbf}[1]{{\bf #1}}  

\newcommand{\nn}{ \nonumber }  


\renewcommand{\arraystretch}{1.5}

\newcommand{\ice}[1]{\relax}

\newcommand{\pivo}[1]{\relax}

\newcommand{\del}[1]{ {\bf \large  [[} #1 {\bf \large  ]]}}

\newcommand{\new}[1]{ {\bf \large  \{ }  #1 {\bf \large  \} } }


\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}

\newcommand{\api}{\frac{\alpha_s}{\pi}}
\newcommand{\apis}{\frac{\alpha_s(s)}{\pi}}
\newcommand{\apiq}{\frac{\alpha_s(Q^2)}{\pi}}
\newcommand{\apimu}{\frac{\alpha_s(\mu^2)}{\pi}}
\newcommand{\apimuo}{\frac{\alpha_s(\mu_0^2)}{\pi}}
\newcommand{\apim}{\frac{\alpha_s(m)}{\pi}}
\newcommand{\apimt}{\frac{\alpha_s(M_{\tau})}{\pi}}
\newcommand{\ba}{\begin{array}} 
\newcommand{\ea}{\end{array}} 
\newcommand{\ds}{\displaystyle} 
\newcommand{\als}{\alpha_s}
\newcommand{\as}{a_s}

\newcommand{\gm}{\gamma_m}
\newcommand{\G}{\Gamma}
\newcommand{\g}{\gamma}
\newcommand{\gaam}{\gamma^{AA}_m}
\newcommand{\gvam}{\gamma^{VV/AA}_m}
\newcommand{\gaaq}{\gamma^{AA}_q}
\newcommand{\gssm}{\gamma^{SS}_m}
\newcommand{\gssq}{\gamma^{SS}_q}
\newcommand{\gppm}{\gamma^{PP}_m}
\newcommand{\gppq}{\gamma^{PP}_q}
\newcommand{\dmu}{\mu^2\frac{d}{d\mu^2}}
\newcommand{\msbar}{\overline{\mbox{MS}}}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\kmu}{\dsp\frac{(m_i(\mu^2)\mp m_j(\mu^2))^2}{Q^2}}
\newcommand{\kq}{\dsp\frac{(m_i(Q^2)\mp m_j(Q^2))^2}{Q^2}}
\newcommand{\kmup}{\dsp\frac{(m_i(\mu^2)\pm m_j(\mu^2))^2}{Q^2}}
\newcommand{\kqp}{\dsp\frac{(m_i(Q^2)\pm m_j(Q^2))^2}{Q^2}}
\newcommand{\pva}{\Pi^{(0)}_{V/A}}
\newcommand{\pvaij}{\Pi^{(0)}_{V/A;ij}}
\newcommand{\pvatr}{\Pi^{(0+1)}_{V/A}}
\newcommand{\pvaijtr}{\Pi^{(0+1)}_{V/A;ij}}
\newcommand{\dvaijz}{\delta^{(D=2)}_{V/A;ij}}
\newcommand{\dvaijv}{\delta^{(D=4)}_{V/A;ij}}
\newcommand{\mtau}{M_{\tau}}
\newcommand{\kk}{\dsp\frac{1}{16\pi^2}}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\dsp}{\displaystyle}
\newcommand{\tst}{\textstyle}
\newcommand{\EQN}{\label}
\newcommand{\ovl}{\overline}
\newcommand{\re}[1]{(\ref{#1})}
\newcommand{\lmq}{l_{\mu Q}}

\newcommand{\BreakO}{ \nonumber \\ &{}&   }

\newcommand{\Break}{ \right. \nonumber \\ &{}& \left. }
\newcommand{\BreakI}{ \right. \nonumber \\ &{}& \left. }
\newcommand{\BreakII}{ \right. \right.  \nonumber \\ &{}& \left.\left. }
\newcommand{\BreakIII}{ \right. \right. \right.  \nonumber \\ &{}& \left.\left.\left. }
\newcommand{\BreakIV}{
\right. \right. \right.\right.  \nonumber \\ &{}& \left. \left.\left.\left.
                    }

\newcommand{\lsb}{\left[}
\newcommand{\rsb}{\right]}




\begin{document}
\begin{thebibliography}{9}

\def\ap#1#2#3{  {Ann.\ Phys.\ B }{\ice{\bf} #1} (#2) #3}
\def\app#1#2#3{  {Act.\ Phys.\ Pol.\ B }{\ice{\bf} #1} (#2) #3}
\def\cpc#1#2#3{  {Comp.\ Phys.\ Commun.\ }{\ice{\bf} #1} (#2) #3}
\def\cmp#1#2#3{  {Comm.\ Math.\ Phys.\ }{\ice{\bf}#1} (#2) #3}
\def\epjc#1#2#3{ {Eur.\ Phys.\ J.\ C }{\ice{\bf} #1} (#2) #3}
\def\fortp#1#2#3{{Fortschr.\ Phys.\ }{\ice{\bf}#1} (#2) #3}
\def\jcp#1#2#3{  {J.\ Comp.\ Phys.\ }{\ice{\bf}#1} (#2) #3}
\def\nima#1#2#3{ {Nucl.\ Inst.\ Meth.\ A }{\ice{\bf} #1} (#2) #3}
\def\npb#1#2#3{  {Nucl.\ Phys.\ B }{\ice{\bf} #1} (#2) #3}
\def\nca#1#2#3{  {Nuovo Cim.\ A }{\ice{\bf} #1} (#2) #3}
\def\plb#1#2#3{  {Phys.\ Lett.\ B }{\ice{\bf} #1} (#2) #3}
\def\prc#1#2#3{  {Phys.\ Reports }{\ice{\bf} #1} (#2) #3}
\def\prd#1#2#3{  {Phys.\ Rev.\ D }{\ice{\bf} #1} (#2) #3}
\def\pR#1#2#3{   {Phys.\ Rev.\ }{\ice{\bf} #1} (#2) #3}
\def\prl#1#2#3{  {Phys.\ Rev.\ Lett.\ }{\ice{\bf} #1} (#2) #3}
\def\pr#1#2#3{   {Phys.\ Reports }{\ice{\bf} #1} (#2) #3}
\def\ppnp#1#2#3{ {Prog.\ Part.\ Nucl.\ Phys.\ }{\ice{\bf} #1} (#2) #3}
\def\sovnp#1#2#3{{Sov.\ J.\ Nucl.\ Phys.\ }{\ice{\bf} #1} (#2) #3}
\def\tmf#1#2#3{  {Teor.\ Mat.\ Fiz.\ }{\ice{\bf} #1} (#2) #3}
\def\yadfiz#1#2#3{{Yad.\ Fiz.\ }{\ice{\bf} #1} (#2) #3}
\def\zpc#1#2#3{  {Z.\ Phys.\ C }{\ice{\bf} #1} (#2) #3}
\def\ibid#1#2#3{ {ibid.\ }{\ice{\bf} #1} (#2) #3}

\bibitem{phys_report}K.G.~Chetyrkin, J.H.~K\"uhn and A.~Kwiatkowski,
Phys.~Rep. 277 (1996) 189.


%\cite{Steinhauser:2002rq}
\bibitem{Steinhauser:2002rq}
For a recent comprehensive  review of the
multiloop calculations see
M.~Steinhauser,
%``Results and techniques of multi-loop calculations,''
Phys.\ Rept.\   364 (2002) 247
[arXiv:hep-ph/0201075].
%%CITATION = HEP-PH 0201075;%%

\bibitem{svz}M.A. Shifman, A.I. Vainshtein, V.I. Zakharov, Nucl. Phys. 
B147 (1979) 385, 448.


\bibitem{OPE:Wilson}K.G.~Wilson, Phys. Rev. 179 (1969) 1499; 
                    D3 (1971) 1818. 

\bibitem{CheKueSte96}
  K.G. Chetyrkin, J.H. K\"uhn, and M. Steinhauser, \plb{371}{1996}{93};
  \npb{482}{1996}{213}; \npb{505}{1997}{40}.

\bibitem{CheHarSte98} K.G. Chetyrkin, R. Harlander, and M. Steinhauser,
  \prd{58}{1998}{014012}.



\bibitem{non_diag}
K.G.~Chetyrkin and M.~Steinhauser,
%``Three-loop non-diagonal current correlators in QCD and NLO corrections  to single top quark production,''
Phys.\ Lett.\ B   502 (2001) 104;
Eur.\ Phys.\ J.\ C  21 (2001) 319.


\bibitem{GorKatLar91SurSam91}
S.G. Gorishny, A.L. Kataev and S.A. Larin,
{Phys. Lett.} { B 259} (1991) 144;\\
L.R. Surguladze and M.A. Samuel,
{Phys. Rev. Lett.} { 66} (1991) 560;
(E) ibid., 2416.


\bibitem{gvvq}
K.G. Chetyrkin, {Phys. Lett.} { B 391} (1997) 402.




\bibitem{gssq}
%Higgs stuff!
%K.G. Chetyrkin, Preprint MPI/PhT/96-61, hep-ph/9608318,
%August 1996.
K.G. Chetyrkin, Phys. Lett. { B 390} (1997) 309.



\bibitem{ChetKuhn90}
K.G. Chetyrkin, J.H. K\"uhn, \plb {248}{1990}{359}.

\bibitem{mq2as3} K.G. Chetyrkin and  J.H. Kuhn,
Phys. Lett. B 406 (1997) 102.

\bibitem{mq4as3}
K.G. Chetyrkin, R.V. Harlander, J.H. K\"uhn, 
Nucl. Phys. B 586 (2000) 56.

\bibitem{TESLA} TESLA, Technical Design Report, DESY 2001-011; ECFA
2001-209; Part III, {\em Physics at an $e^+e^-$ Li\-near Collider},
R.-D. Heuer, D. Miller, F. Richard, P. Zerwas, eds.



\bibitem{KniKue90b} B.A.~Kniehl, J.H.~K\"uhn, \plb{224}{1990}{229};
 \npb{329}{1990}{547}.

  

\bibitem{CheKue93} K.G. Chetyrkin and J.H. K\"uhn, \plb{308}{1993}{127}.
  

\bibitem{CheTar94LarRitVer94} 
  K.G. Chetyrkin and O.V. Tarasov,
  \plb{327}{1994}{114}; \\
  S.A. Larin, T. van Ritbergen and J.A.M. Vermaseren,
  \plb{320}{1994}{159}.


  

\bibitem{msbar} W.A. Bardeen, A.J. Buras, D.W. Duke and
T. Muta, Phys. \ Rev. D \textbf{18}, 3998  (1978).


\bibitem{VV:renormalons}
M.~Beneke, \npb{405}{1993}{424}.


%\cite{Baikov:2001aa}
\bibitem{Baikov:2001aa}
P.A.~Baikov, K.G.~Chetyrkin and J.H.~K\"uhn,
%``The cross section of e+ e- annihilation into hadrons of order  alpha(s)**4 n(f)**2
%in perturbative QCD,''
Phys.\ Rev.\ Lett.\   88, (2002)  012001 .


 

\bibitem{KS} A.L.~Kataev and V.V.~Starshenko,
Mod. Phys. Lett.  {A 10}, 235 (1995).


\bibitem{BKCh:tau}
P.A.~Baikov, K.G.~Chetyrkin and J.H.~K\"uhn, 
``Towards Order   $\alpha_s^4$ Accuracy in  $\tau$-decays'',
hep-ph/0212299.


\bibitem{Chetyrkin:uw}
K.G.~Chetyrkin,
%R(s)in pQCD,
%\cite{Chetyrkin:uw}
Nucl.\ Phys.\ Proc.\ Suppl.\   89 (2000) 47.
%%CITATION = NUPHZ,89,47;%%



\bibitem{beta3loop}
O.V. Tarasov, A.A. Vladimirov, A.Yu. Zharkov,
Phys.\ Lett.\ B 93 (1980) 429;
S.A. Larin and J.A.M. Vermaseren,
Phys. Lett. B 303 (1993) 334.

\bibitem{gamma3loop}
O.V. Tarasov, preprint JINR P2-82-900 (1982).


\bibitem{GorKatLar86}
S.G. Gorishny, A.L. Kataev, S.A. Larin,
Nouvo~Cim. 92A (1986) 119.

\bibitem{RitVerLar97}
  T. van Ritbergen, J.A.M. Vermaseren, and S.A. Larin \plb{400}{1997}{379}.

\bibitem{Che97}
K.G. Chetyrkin, \plb{404}{1997}{161}.

\bibitem{LarRitVer972}
  S.A. Larin, T. van Ritbergen, and J.A.M. Vermaseren, \plb{405}{1997}{327}.


\bibitem{bai1} %
P.\,A.\,Baikov, Phys.\,Lett.\,  B 385, (1996) 404;
Nucl.~Instr.~\& Methods~ A 389 (1997) 347;
P.\,A.\,Baikov, M. Steinhauser, Comput.\,Phys.\,Commun. 115 (1998)
161.

\bibitem{bai2} % criterion
P.\,A.\,Baikov, Phys.\,Lett.\,  B474 (2000) 385-388;





%\cite{Fliegner:2000uy}
\bibitem{Fliegner:2000uy}
D.~Fliegner, A.~Retey, and J.A.~Vermaseren,
%``Parallelizing the symbolic manipulation program FORM.  I: Workstation
%clusters and mes sage passing,''
hep-ph/0007221.
%%CITATION = HEP-PH 0007221;%%

\bibitem{Vermaseren}
J.A.M.~Vermaseren,
%"New features of FORM"
math-ph/0010025.



\bibitem{ste}
P.M.~Stevenson,
Phys. Rev.  D  {23}, 2916 (1981);
Phys. Lett. B {100}, 61 (1981);
Nucl. Phys. B {203}, 472 (1982);
Phys. Lett. B {231}, 65 (1984).


\bibitem{gru}
G.~ Grunberg, Phys. Lett.  B {95}, 70 (1980);
Phys. Rev. D  {29}, 2315 (1984).

\bibitem{krasn}
N.V.~Krasnikov, Nucl. Phys. B {192}, 497 (1981);



%\cite{Kataev:1994wd}
\bibitem{Kataev:1994wd}
A.L.~Kataev and V.V.~Starshenko,
%``Estimates of the higher order QCD corrections: Theory and applications,''
Nucl.\ Phys.\ Proc.\ Suppl.\   39BC , 312 (1995),
 [arXiv:hep-ph/9408395].
%%CITATION = HEP-PH 9408395;%%





\bibitem{BG}  D.J.~Broadhurst and A.G.~Grozin,
Phys. Rev. D  \textbf{52}, 4082 (1995); for a recent 
discussion see also \cite{BK}.


\bibitem{BK}  D.J.~Broadhurst and A.L.~Kataev,
hep-ph/0207261.




\bibitem{CheKniSir97}
 K~G.~Chetyrkin, B.A.~Kniehl and A.~Sirlin, \plb{402}{1997}{359}.





\end{thebibliography}


\begin{titlepage}
\noindent
%
% Datum
%
\hfill THEP 02/18 \\
\mbox{}
\hfill TTP02--45\\
\mbox{}
%\hfill  December  2002  \\   
\mbox{}
\hfill hep-ph/0212303 \\
\hfill {}

\vspace{0.5cm}
\begin{center}
  \begin{Large}
  \begin{bf}
Five-loop vacuum polarization in pQCD:
$\mathcal{O}(m_q^2\alpha_s^4 n_f^2)$ contribution 
  \end{bf}
  \end{Large}

%
% Author
%
  \vspace{0.8cm}

  \begin{large}
P.A.~Baikov\\
 {\small {\em 
Institute of Nuclear Physics, 
Moscow State University
Moscow~119992, Russia}}\\
K.G. Chetyrkin\footnote{On leave from Institute for Nuclear Research
of the Russian Academy of Sciences, Moscow, 117312, Russia.} \\
{\small {\em Fakult{\"a}t f{\"u}r Physik
 Albert-Ludwigs-Universit{\"a}t Freiburg,
D-79104 Freiburg, Germany }}
J.H.~K\"uhn  \\[3mm]
{\small {\em
    Institut f\"ur Theoretische Teilchenphysik,
    Universit\"at Karlsruhe,
     D-76128 Karlsruhe, Germany}
}
%[-5mm]
\end{large}

\vspace{0.8cm}
{\bf Abstract}
\end{center}
\begin{quotation}
\noindent

We present the analytical calculation of the contribution of order
$m_q^2 \alpha_s^4 n_f^2$ to the absorptive part of the vacuum
polarization function of vector currents. This term constitutes an
important gauge-invariant part of the full $\mathcal{O}(m_q^2 \als^4)$
correction to the total cross-section of $e^+ e^-$ annihilation into
hadrons.  The results are 
compared to predictions
following from various optimization schemes.


\end{quotation}
\end{titlepage}




\section{Introduction}

In QCD the correlator of two currents is a central object from which
important physical consequences can be deduced (for a detailed review
see, e.g.~\cite{phys_report}).  In particular, important physical
observables like the cross-section of $e^+ e^-$ annihilation into
hadrons and the decay rate of the $Z$ boson are related to the vector
and axial-vector current correlators.  Furthermore, total decay rates
of CP even or CP odd Higgs bosons can be obtained by considering the
scalar and pseudo-scalar current densities, respectively.

From the theoretical viewpoint the two-point correlators are
especially suited for  evaluations
in the  framework of perturbative QCD (pQCD)
\cite{Steinhauser:2002rq}. Indeed, due to the simple kinematics (only one
external momentum) even multiloop calculations can be analytically
performed. Non-perturbative contributions can be
effectively  controlled through the  the operator product
expansion \cite{svz,OPE:Wilson}.  As a consequence, the results for
practically all physically interesting correlators (vector,
axial-vector, scalar and pseudo-scalar) are available up to order
$\alpha_s^2$ taking into account the {\em full} quark mass dependence
\cite{CheKueSte96,CheHarSte98,non_diag}.

In many important cases (with the Z decay rate  a prominent
example) the external momentum  is much larger than the masses of
the (active) quarks involved. This 
justifies to neglect   these masses in a first approximation 
which significantly simplifies the calculation.
As a result, in massless QCD the vector and scalar correlators are
analytically known to $\als^3$ \cite{GorKatLar91SurSam91,gvvq,gssq}.
The residual quarks mass effects can be taken into account via the
expansion in quark masses.  At present this has been done for the
quadratic and quartic terms to the same $\als^3$ order
\cite{ChetKuhn90,mq2as3,mq4as3}.


During the past years, in particular through the analysis of $Z$
decays at LEP and of $\tau$ decays, an enormous reduction of the
experimental uncertainty (down to $\mathcal{O}(10^{-3})$) and  in $R(s)$ has been achieved
with the perspective of a further reduction by a factor of four at a
future linear collider \cite{TESLA}.

Inclusion of the $\mathcal{O}(\als^3)$ corrections
\cite{GorKatLar91SurSam91} is  mandatory already now. Quark mass
effects as well as corrections specific to the axial current
\cite{KniKue90b,CheKue93,CheTar94LarRitVer94} must be included for the case of $Z$-decays.  The
remaining theoretical uncertainty from uncalculated higher orders is
at present comparable to the experimental one
\cite{phys_report}. 
Thus, the full calculation of the next  contributions,
those of $\mathcal{O}(\als^4)$, 
to $R(s)$ is on agenda.

In   massless approximation $R(s)$ is conveniently written as
\bea
R(s) &=& \sum_{f} Q_f^2\, 3 \left( 
\rule{0mm}{7mm}
\right.
1 +
\as + \as^2  (1.98571 - 0.115295 \, n_f) + 
\nn
\\
&{}& 
\as^3  (-6.63694 - 1.20013 \, n_f - 0.00517836 \, n_f^2 ) + 
 \as^4 \,  r^{V,4}_0
\left.
\rule{0mm}{7mm}
\right)
\dots	
{},
\label{R(s):nf:as^3+rV4}
\eea
where $\as = \alpha_s(\mu^2=s)/\pi$ and the standard $\msbar$ renormalization
prescription \cite{msbar} is understood.
The $\as^4$ term can be further decomposed as 
a polynomial in $\, n_f$, namely
\beq
r^{V,4}_0 = r^{V,4}_{0,0} + r^{V,4}_{0,1} \, n_f + r^{V,4}_{0,2} \, n_f^2
+ r^{V,4}_{0,3} \, n_f^3 
{}.
\label{rV:nf expansion}
\eeq
The term of order $\als^4 \, n_f^3$ (and, in fact, all terms of order
$\als(\als \, n_f)^n$) have been obtained earlier by summing the
renormalon chains \cite{VV:renormalons}. These are  technically very simple
to compute\footnote{At least for a fixed number of loops.}, 
numerically small ($r^{V,4}_{0,3} = 0.02152$) and not directly sensitive to the nonabelian
character of QCD, in contrast to the terms of order $\als^4 n_f^2$. 
The first nontrivial contribution --- 
the subleading ${\mathcal{O}}(n_f^2)$  piece in (\ref{rV:nf expansion}) ---
have been only recently analytically  computed in \cite{Baikov:2001aa} with the 
result\footnote{For brevity we display it in the numerical form.}
\beq
r^{V,4}_{0,2} = -0.7974
{}.
\label{rV4nf2:result}
\eeq


Taken by itself eq.~(\ref{rV4nf2:result}) is  not of much use for
the phenomenology for obvious reasons. However, 
it gives a  strong  extra support to the well-known  prediction
of the full ${\mathcal{O}}(\alpha_s^4)$ contribution  
of the Kataev and Starshenko\cite{KS} (for a detailed discussion see 
\cite{BKCh:tau}).
The prediction is often used to estimate the theoretical uncertainty 
in $R(s)$ due to higher order not yet computed  perturbative
contributions.  

Another point of concern are the quark mass effects at order
$\alpha_s^4$. Fortunately, simple estimates (see
e.g. \cite{ChetKuhn90}) show that, say, the quadratic mass effects are
completely under control\footnote{ \emph{Provided} one uses the
properly chosen parameterization in terms of the running quark mass.}
at the scale of the Z-boson mass since terms of order
${\mathcal{O}}(\alpha_s^3)$ are available and small. This holds true
for the vector correlator with the leading term proportional 
$\alpha_s m_b^2$ as well as for the axial vector correlator, where the the
leading in $\alpha_s$ term is present already at the Born level and
thus proportional to $m_b^2$.  Quarks mass effect are getting
progressively more important at lower energies.  From the conceptual
viewpoint it
 would be  important  to test the evolution of the
strong coupling as predicted by the beta function and the 
standard QCD matching procedure through a
determination of $\as$ from essentially the same observable, however,
at lower energy. The region from several GeV above charm threshold
(corresponding to the maximal energy of BEPC around $5.0$ GeV) to just
below the $B$ meson threshold at around $10.5$ GeV corresponding to
the ``off resonance" measurements of CESR or  B-meson factory is  particularly suited
for this purpose.  As a consequence of the favorable error
propagation, 
\[
\delta \as(s) = \frac{\as^2(s)}{\as^2(M_Z^2)} \delta \as(M_Z^2)
{},
\]
the accuracy in the measurement (compared to $91$ GeV)
may decrease by factor of about 3 or even 4 at 10 and 5.6 GeV
respectively, to achieve comparable precision in 
$\La_{QCD}$. 



Technically speaking, the evaluation of the leading in $n_f$ term of
order $O(m_q^2\alpha_s^4 n_f^3)$ is again rather simple while the
subleading term of order $O(m_q^2\alpha_s^4 n_f^2)$ presents a problem
comparable to that in the massless limit.  In this work we describe
the corresponding calculation and its results.  We limit the
discussion to the vector current correlator, which is relevant for
hadron production through the electromagnetic current. We also compare
our results with predictions following from various optimization
schemes.  Our basic conclusion is once again that the FAC/PMS
optimization predictions are remarkably close to reality. They
provides a quantitative argument in favour of the corresponding full
prediction.


\section{Generalities}
To fix notation we start  considering  two-point correlator
of vector  currents and the corresponding vacuum polarization
function  ($j_\mu^v= \ovl{Q}\gamma_\mu Q$;
$Q$ is a  quark field with  mass $m$, all other $n_f-1$ quarks  are
assumed to be massless)
\begin{eqnarray}
\Pi_{\mu\nu}(q)  &=&
 i \int {\rm d} x e^{iqx}
\langle 0|T[ \;
\;j_{\mu}^{v}(x)j_{\nu}^{v}(0)\;]|0 \rangle
=
\displaystyle
(-g_{\mu\nu}q^2  + q_{\mu}q_{\nu} )\Pi(q^2)
{}.
\label{PiV}
\end{eqnarray}
The  physical observable $R(s)$ is related to $\Pi(q^2)$
by
\beq
R(s) = 12 \pi \Im \, \Pi(q^2 + i\epsilon)
{}.
\label{R(s):def}
\eeq
It is convenient to decompose $R(s)$ into the  massless and the quadratic
terms
\bea
R(s) &=& 3 \left\{r^V_0 +  \frac{m^2}{s} r^{V}_2 \right\}+\dots 
= 3\left\{\sum_{i \ge 0} a_s^i\left(
 r^{V,i}_0 + \frac{m^2}{s} r^{V,i}_2 \right)
                                    \right\} +\dots
\nn
{}.
\eea
Here we have set the normalization scale  $\mu^2= s$;
$a_s = \als(s)/\pi$ and dots stand for corrections proportional
to $m^4$ and higher powers of the quark mass.


For the calculation of $r^{V,4}_0$ we  had to deal with divergent
parts of five-loop diagrams and finite parts of the four-loop ones
(see, e.g the corresponding discussion in
\cite{gvvq,Chetyrkin:uw}). 
In fact, as was first discovered in
Ref.~\cite{ChetKuhn90}, the quadratic mass correction $r^{V,4}_2$ can
be obtained with the help of renormalization group methods {\em
exclusively} from the four-loop function $\Pi_2$ defined as
($ a_s = \als(\mu^2)$ )
%,\mu = -q^2$  
\[
\nn
\Pi   = 
\frac{3}{16 \pi^2} \left(
\Pi_0  + \frac{m^2}{-q^2} \Pi_2
\right)
{},
\ \ \ 
\Pi_2 = \sum_{i \ge 0}\   a_s^i k^{V,i}_2
{}.
\]
Indeed, due to an extra factor of $m^2$ the function $\Pi_2$  obeys an
\emph{unsubtracted} dispersion relation. As a result the following
\emph{homogeneous} RG equation  holds:
\beq
\left(
\mu^2\frac{\partial}{\partial\mu^2}
 +
\g_m(\as) m \frac{\partial}{\partial m}
 +
\beta(a_s)
a_s
\frac{\partial}{\partial a_s}
\right)
      \Pi_2
          = 0
\label{rg:Pi2}
{}.
\eeq
Equivalently, since $\Pi_2$ is independent of $m$, 
\beq
\frac{\prd }{\prd L} \Pi_2 =
\left(
-2\g_m(\as)
-
 \beta( a_s) a_s\frac{\prd }{\prd a_s}
\right)
\Pi_2
\label{rg:Pi2:L}
{},
\eeq
where  $L = \ln\frac{\mu^2}{-q^2}$.
Once the rhs of eq.~(\ref{rg:Pi2:L}) is  known, one can  find  the function 
$\Pi_2$ up to a constant contribution which  has no effect   on $R(s)$.  
Thus, to compute, say, 
the  $\mathcal{O}(\frac{m^2}{s} \as^3)$ correction to $R(s)$ one  needs to know
\begin{itemize}
\item
The three-loop QCD $\beta$ function
function  (starting at ${\mathcal{O}} (\alpha_s^2)$) and 
the quark mass anomalous dimension (starting at ${\mathcal{O}}(\alpha_s)$):
available from    \cite{beta3loop,gamma3loop}, each 
up to order  $\alpha_s^3$.
\item
The very polarization operator  $\Pi_2$ at {\em three} loops
(that is  of order  $\alpha_s^2$):
was computed also long ago in \cite{GorKatLar86}.
\end{itemize}
Altogether, this implies that the ${\cal O}(\frac{m^2}{s} \as^3)$  term in $R(s)$ 
\emph{could} have been been computed as early as in 1986. But in reality this  
was done only by 5 years later \cite{mq2as3}.

From the above discussion it is clear which ingredients are
necessary to compute quadratic  mass terms in $R(s)$ at ${\cal O}(\as^4)$:
\begin{enumerate}
\item
The four -loop QCD $\beta$ function
function and quark mass anomalous dimension:
available  from \cite{RitVerLar97,Che97,LarRitVer972}.
\item
The  polarization operator $\Pi_2$ at {\emph four} loops,
including its  {\em constant} part.
\end{enumerate}


\section{Results}

Using the new technique described in \cite{bai1,bai2,Baikov:2001aa}
and the parallel version of FORM \cite{Fliegner:2000uy,Vermaseren},
we have computed the leading and subleading (in $n_f$) contributions of order 
$\als^3$   to the polarization function $\Pi_2$
($\mathcal{O}(\alpha_s^2)$ results are known from \cite{GorKatLar86})
%The coefficients  $k^{V,i}_2$ for $i=1,2$ are known from \cite{GorKatLar86}
%while  new result for  $k^{V,3}_2$ reads
\bea
 \Pi_{2} = -8 &-&\frac{64}{3}\, a_s + a_s^2 \left\{
\frac{95}{9} \,n_f +-\frac{18923}{54}-\frac{784}{27} \,\zeta_{3}+\frac{4180}{
27} \,\zeta_{5}\right\}
\nonumber
 \\ &+& a_s^3 \left\{
 \lsb -\frac{5161}{1458}-\frac{8}{27} \,\zeta_{3} \rsb  \, n_f^2 +\lsb \frac{
62893}{162}+\frac{424}{27} \,\zeta_3^2-\frac{4150}{243} \,\zeta_{3}\BreakII
 \phantom{ a_s^3 ||}
+\frac{20}{3} \,\zeta_{4}-\frac{28880}{243} \,\zeta_{5} \rsb  \,n_f +k^{[V]3}
_{2,0}\right\} 
{}
\label{Pi2:nf} \end{eqnarray}
or, numerically,
\beq
\Pi_2 =
-  8 - 21.333 \, a_s 
+  a_s^2 \, ( 10.56 \, n_f -224.80 ) 
+ a_s^3\, (- 3.896 \, n_f^2 +  274.37\, n_f   + k^{V,3}_{2,0} )
{},
\label{Pi_2:num}
\eeq
where we have  set  $a_s = \alpha_s(-q^2), \mu^2 = -q^2$. 

The  straightforward  use of eqs.~(\ref{R(s):def},\ref{rg:Pi2:L}) yields 
($a_s = \alpha_s(s), \mu^2 = s$)
\bea r^{V}_{2} = 12 \frac{\alpha_s}{\pi} &+& a_s^2 \left\{
-\frac{13}{3} \,n_f +\frac{253}{2}\right\}
\nonumber
 \\ &+& a_s^3 \left\{
 \lsb \frac{125}{54}-\frac{1}{9} \pi^2 \rsb  \, n_f^2 +\lsb -\frac{4846}{27}+
\frac{17}{3} \pi^2-\frac{466}{27} \,\zeta_{3}\BreakII
 \phantom{ a_s^3 ||}
+\frac{1045}{27} \,\zeta_{5} \rsb  \,n_f +2442-\frac{285}{4} \pi^2+\frac{490}
{3} \,\zeta_{3}-\frac{5225}{6} \,\zeta_{5}\right\}
\nonumber
 \\ &+&a_s^4\left\{
 \lsb -\frac{2705}{1944}+\frac{13}{108} \pi^2 \rsb  \, n_f^3 +\lsb \frac{9194
3}{486}-\frac{1121}{108} \pi^2+\frac{53}{9} \,\zeta_3^2\BreakII
 \phantom{a_s^4||}
+\frac{13}{324} \,\zeta_{3}-\frac{3610}{81} \,\zeta_{5} \rsb  \, n_f^2 + r_{2
,0}^{V,4} \,n_f +r_{2,1}^{V,4}\right\} {}.
\label{rV4_2V:nf} \end{eqnarray}
Numerically
\bea
r^V_2 = {}
12\,  a_s &+&   a_s^2 \left(-4.3333 \ n_f+  126.5  \right)
+
a_s^3
\left(
 1.2182 \, n_f^2- 104.167  \, n_f  + 1032.14 
\right)
\nn
\\
&+& a_s^4
\left(
- 0.20345 \,  n_f^3
+ 49.0839  \,n_f^2  
+ \  r^{V,4}_{2,1}\, n_f \ + r^{V,4}_{2,0}
\right)
{}.
\label{rV_2_2:num}
\eea

\section{Discussion}
Let us contrast eq.~(\ref{Pi2:nf}) with predictions based on the
optimization methods FAC (Fastest Apparent Convergence) and PMS
(Principle of Minimal Sensitivity)
\cite{ste,gru,krasn,KS}. As is well known in many cases FAC and PMS
predictions are quite close \cite{Kataev:1994wd}. To compare with an
independent  
prediction   
%check 
we will also use so-called NNA (Naive
NonAbelianization) approach of \cite{BG}.




In Table 1 we list the FAC and PMS predictions for the 
function $\Pi_2$ 
(see $\leftarrow$ \cite{CheKniSir97}). 
{}From the three entries corresponding to $n_f = 3,4$ and $5$ one
easily restores the FAC/PMS prediction for the $n_f$ dependence of the
$\as^3$ term in $\Pi_{2}$ 
\bea 
k^{V,3}_{2}(\mathrm{FAC}) &=& -4.2 \, n_f^2 + 277 \, n_f -2886    
\label{K4:FAC:nf}
{},
\\
k^{V,3}_{2}(\mathrm{PMS}) &=& -4.2 \, n_f^2   + 282 \, n_f   -2916
\label{K4:PMS:nf}
{}.
\eea
The comparison with  (\ref{Pi_2:num}) shows  very good agreement with
the calculated  terms of order $n_f$ and $n_f^2$.  As a natural next
step we will now use the available \emph{exact} information about the
coefficients $k^{V,3}_{2,2}$ and $k^{V,3}_{2,1}$ to find
$k^{V,3}_{2,0}$ from FAC/PMS prediction for a selected value of $n_f=
3$. The choice seems to be natural as in many cases FAC/PMS
predictions made for $n_f= 3$ are in better agreement with the exact
results (see. e.g. \cite{KS,Kataev:1994wd,BKCh:tau}).  We
obtain (in fact, very close values are produced with $n_f =4$ and 5)
\beq
k^{V,3}_{2,0}(\mathrm{FAC},nf = 3) =  -2880, \   \ \ k^{V,3}_{2,0}(\mathrm{PMS},nf = 3) =  -2897
{}.
\label{FAC/PMS:nf:3}
\eeq

Now we turn to NNA. The corresponding prescription is to simply
make the substitution  $n_f \to -3/2\beta_0 = n_f -33/2$ in  the 
leading term of the $n_f$ expansion. The result is
\beq 
k^{V,3}_{2}(\mathrm{NNA})  =  -1060.69  + 128.568 \, n_f  - 3.896 \, n_f^2  
{}.
\label{NNA}
\eeq
This  prediction is in obvious conflict  with  the
result of  the direct calculation of the term linear in $n_f$.




\begin{table}[hbt]
\caption{\label{t1}
Estimates for  the  coefficient $k^{V,3}_2$
based on  FAC and PMS optimization schemes.
}
\renewcommand{\arraystretch}{1.2}
\begin{center}
\begin{tabular}{|c|c|c|c|}
\hline
Method& $n_f=3$ & $n_f=4$ & $n_f=5$  \\
\hline\hline
FAC & $-2092  $  & $ - 1844 $ & $- 1604 $ \\
\hline
PMS & $ -2109 $ & $-1856$ & $-1612 $\\
\hline
\end{tabular}
\end{center}
\end{table}



\section{Conclusions}

We have presented the analytical calculation of contributions of
order $m_q^2 \alpha_s^4 n_f^2$ to the vacuum polarization function of
vector currents. Predictions of FAC, PMS- and NNA optimization methods
are tested against the exact results.  Good quantitative agreement
is found for the first two cases.  NNA  
seemingly  predicts only sign and  order of magnitude
of analytical results.



The authors are grateful to M.~Steinhauser   
and R.~Harlander for the careful reading 
of the manuscript and   very useful comments.  This work was supported by the
DFG-Forschergruppe {\it ``Quantenfeldtheorie, Computeralgebra und
Monte-Carlo-Simulation''} (contract FOR 264/2-1), by INTAS (grant
00-00313), by RFBR (grant 01-02-16171), by Volkswagen Foundation and
by the European Union under contract HPRN-CT-2000-00149.




\end{document}


